Author's response to reviews

Title: "Taking Your Place at the Table": An Autoethnographic Study of Chaplains' Participation on an Interdisciplinary Research Team

Authors:

Allison Kestenbaum (alkestenbaum@jtsa.edu)
Jennifer James (jennifer.james@ucsf.edu)
Stefana Morgan (stefana.morgan@ucsf.edu)
Michele Shields (michele.shields@ucsf.edu)
Will Hocker (will.hocker@ucsfmedctr.org)
Michael Rabow (mrabow@medicine.ucsf.edu)
Laura B. Dunn (laura.dunn@ucsf.edu)

Version: 4
Date: 24 February 2015

Author's response to reviews:

Manuscript Title: “Taking Your Place at the Table”: An Autoethnographic Study of Chaplains’ Participation on an Interdisciplinary Research Team

We greatly appreciate the reviewers’ time and effort in reviewing this manuscript. The reviewers indicated that there are no further revisions required.

We have made the additions and revisions requested in the Editorial Request. Responses to each of the points in the Editorial Request are detailed below.

1. Title page - The title page now contains all of this information.

2. Competing Interests - We added a Competing interests section between the Conclusions and Authors’ contributions.

3. Acknowledgements - We included an Acknowledgements section between the Authors contributions section and Reference list.

4. Requesting name of ethics committee - The name of the committee is the UCSF Committee on Human Research. We added this to the Methods section.

5. Informed consent - All research participants and research team members provided written informed consent for this study. We have added this to the Methods section.